Author's response to reviews

Title: Perceptions of young Jordanian adults to proposed anti-tobacco pictorial warning labels

Authors:

Feras I Hawari (fhawari@khcc.jo)
Rasha K Bader (rbader@khcc.jo)
Hamza M Beano (hamza.beano@hotmail.com)
Nour A Obeidat (nobeidat@khcc.jo)
Hiba S Ayub (hayub@khcc.jo)
Malek A Habashneh (malek_131@yahoo.com)
Aisha S Shtaiwi (ashtaiwi@khcc.jo)
Rawan A Shihab (rshihab@khcc.jo)
Hala N Madanat (hmadanat@mail.sdsu.edu)
Thomas E Novotny (tnovotny@mail.sdsu.edu)

Version: 6 Date: 7 May 2011

Author's response to reviews:

To the editorial production team:

Please find below our response to comments received May 6, 2011 on our submission to BMC Public Health (MS 1013641144475992, “Perceptions of young Jordanian adults to proposed anti-tobacco pictorial warning labels”).

1- Additional files: We have added a section as detailed by your comments to address additional files. Where appropriate within the text, that is whenever we describe the instrument or its content, we have referred to additional files. We have accordingly renamed files and uploaded the newer version.

2- References: We have modified the style of listing references as requested.

We would like to thank you for the opportunity to further improve our submission. We hope the changes address all points raised sufficiently.

Regards,
Feras Hawari, MD